Author's response to reviews

Title: Developing a web-based information resource for palliative care: an action-research inspired approach

Authors:

Annette F Street (A.Street@latrobe.edu.au)
Kathleen Swift (K.Swift@latrobe.edu.au)
Merilyn Annells (M.Annells@latrobe.edu.au)
Roger Woodruff (Roger.Woodruff@austin.org.au)
Terry Gliddon (T.Gliddon@royalnursing.org.au)
Anne Oakley (A.Oakley@mcm.org.au)
Goetz Ottmann (G.Ottmann@latrobe.edu.au)

Version: 4 Date: 27 August 2007

Author's response to reviews:

Dear Editor,

I am uploading a revised version of the manuscript entitled "Developing a web-based information resource for palliative care: an action-research inspired approach"

The version includes changes to the format as requested in your email from 20 August.

Regards,

Goetz Ottmann